Variation in Genome-Wide Levels of Meiotic Recombination Is Established at the Onset of Prophase in Mammalian Males

Brian Baier¹, Patricia Hunt¹, Karl W. Broman², Terry Hassold¹*

1 School of Molecular Biosciences, Washington State University, Pullman, Washington, United States of America, 2 Department of Biostatistics and Medical Informatics, University of Wisconsin-Madison, Madison, Wisconsin, United States of America

Abstract

Segregation of chromosomes during the first meiotic division relies on crossovers established during prophase. Although crossovers are strictly regulated so that at least one occurs per chromosome, individual variation in crossover levels is not uncommon. In an analysis of different inbred strains of male mice, we identified among-strain variation in the number of foci for the crossover-associated protein MLH1. We report studies of strains with "low" (CAST/EiJ), "medium" (C3H/HeJ), and "high" (C57BL/6J) genome-wide MLH1 values to define factors responsible for this variation. We utilized immunofluorescence to analyze the number and distribution of proteins that function at different stages in the recombination pathway: RAD51 and DMC1, strand invasion proteins acting shortly after double-strand break (DSB) formation, MSH4, part of the complex stabilizing double Holliday junctions, and the Bloom helicase BLM, thought to have anti-crossover activity. For each protein, we identified strain-specific differences that mirrored the results for MLH1; i.e., CAST/EiJ mice had the lowest values, C3H/HeJ mice intermediate values, and C57BL/6J mice the highest values. This indicates that differences in the numbers of DSBs (as identified by RAD51 and DMC1) are translated into differences in the number of crossovers, suggesting that variation in crossover levels is established by the time of DSB formation. However, DSBs per se are unlikely to be the primary determinant, since allelic variation for the DSB-inducing locus Spo11 resulted in differences in the numbers of DSBs but not the number of MLH1 foci. Instead, chromatin conformation appears to be a more important contributor, since analysis of synaptonemal complex length and DNA loop size also identified consistent strain-specific differences; i.e., crossover frequency increased with synaptonemal complex length and was inversely related to chromatin loop size. This indicates a relationship between recombination and chromatin compaction that may develop as DSBs form or earlier during establishment of the meiotic axis.

Citation: Baier B, Hunt P, Broman KW, Hassold T (2014) Variation in Genome-Wide Levels of Meiotic Recombination Is Established at the Onset of Prophase in Mammalian Males. PLoS Genet 10(1): e1004125. doi:10.1371/journal.pgen.1004125

Editor: Molly Przeworski, University of Chicago, United States of America

Received October 3, 2013; Accepted December 4, 2013; Published January 30, 2014

Copyright: © 2014 Baier et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License., which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was supported by NIH grants R01 HD21341 (to TH) and R01 ES013527 (to PH). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: terryhassold@wsu.edu

Introduction

Recombination is a defining event of meiosis, resulting in the physical exchange of DNA between homologous chromosomes. It is generally thought that this is essential for proper alignment and subsequent segregation of homologs during meiosis I and, indeed, evidence from yeast [1,2], Caenorhabditis elegans [3], Drosophila melanogaster [4], and mammals [5] indicates that alterations in the number or positioning of recombination events increase the likelihood of meiotic nondisjunction. For this reason, it might be expected that recombination levels are strictly regulated but, surprisingly, substantial inter-individual variation in recombination is observed in most mammalian species. For example, linkage studies have demonstrated extensive variation in recombination rates and/or recombination hotspot usage in human males and females (e.g., [6,7,8]), and cytological studies of recombination indicate 15-25% individual differences in genome-wide recombination levels in rhesus and human males [9,10], and similar levels of variation in different inbred strains of mice [11,12]. Further, the frequency and location of exchanges vary between the sexes; e.g., in humans the female genetic map is approximately 1.6 fold longer than that of males, with interstitial exchanges being more common in females than males (e.g., [13,14,15]). These differences beg an obvious question: what is responsible for the variation in recombination levels observed among individuals or inbred strains?

Arguably, there are at least three different time-points in the recombination process at which variation in recombination levels could arise. First, variation could be induced at the beginning of the recombination pathway, when double-strand breaks (DSBs) are formed. For example, some individuals might have a greater number of DSBs than others and, assuming that similar proportions of DSBs are converted into crossovers, the end result would be variation in the number of crossovers. On the surface, recent studies in mice would seem to eliminate this possibility: as in other model organisms [16], crossover homeostasis operates in male mice to ensure the presence of a minimum number of crossovers over a wide range of DSBs, suggesting that variation in DSBs does not translate into variation in crossovers [17]. However, these analyses were intended to assess variation in recombination among mice with different numbers of functional

Author Summary

During prophase of meiosis, homologous chromosomes exchange genetic material, in a process known as crossingover. Crossovers are thought to be essential for proper separation of chromosomes during meiosis but, surprisingly, most mammalian species exhibit substantial individual variation in the number of crossovers per cell. We investigated the basis for this variation by examining localization patterns of crossover-associated proteins in inbred strains of male mice with differing average numbers of crossovers per spermatocyte. Our results indicate that the strain-specific variation is established early in meiotic prophase, possibly even before the DNA is broken in advance of subsequent exchanges between homologous chromosomes.

alleles for the DSB-inducing locus Spol1, and do not preclude the possibility that genetic background differences may influence the overall number of crossovers. Second, variation could arise at some stage in the processing of recombination intermediates. For example, genotypic differences at RNF212, encoding a protein with homology to crossover promoting proteins in S. cerevisiae and C. elegans, is the best characterized determinant of individual variation in genome-wide recombination rates in humans (e.g., [18]). Recently, analyses of mice deficient for RNF212 indicate that it acts downstream of the initiation of DSBs, stabilizing joint DNA molecules and promoting the resolution of DSBs as crossovers [19]. Taken together, these observations provide evidence that events occurring after the formation DSBs can, indeed, affect the eventual number of recombination events, although it is not clear that this accounts for all of the amongindividual or among-strain differences that have been reported in mammals. Third, differences might arise at the end of the recombination pathway, owing to individual variation in proteins such as MLH1 and MLH3 that are important in the resolution of double Holliday junctions into crossovers (e.g., [20]).

To discriminate among these possibilities in a mammalian system, we took advantage of strain-specific differences in genomewide rates of meiotic recombination in the male mouse [11]. Specifically, we used immunofluorescence to examine the localization patterns of early-, mid-, and late-acting meiotic recombination proteins, asking whether the patterns were the same or different among inbred strains known to have "low", "mid", or "high" genome-wide rates of recombination. For each strain, we analyzed the number of foci for RAD51 and DMC1 (strand invasion proteins acting shortly after double-strand break formation; [21]), MSH4 (part of the complex stabilizing double Holliday junctions; [22]), and BLM (thought to have anti-crossover activity; e.g., see [23]), and compared these observations with results of analyses of the CO-associated protein MLH1 [20].

Our results demonstrate that inter-strain differences in crossovers (MLH1 foci) are preceded by proportionally similar differences in early-acting recombination proteins, indicating that the variation is established at, or before, the time of DSB formation. Subsequent analyses of males heterozygous for the gene encoding the DSB-inducing protein SPO11 [24,25] allowed us to eliminate DSBs per se as the source of the variation, since *Spo11* heterozygotes exhibited a decrease in DSBs, but not in MLH1 foci.

In analyses of chromatin loop size and synaptonemal complex (SC) length, we detected striking differences among the three inbred strains, but not between *Spo11* heterozygotes and their wildtype littermates. Taken together with the observations on

recombination proteins, our results suggest that strain-specific differences in chromatin architecture, presumably established prior to the initiation of recombination, are important determinants of variation in crossover frequency.

Results

Strain-specific variation in MLH1 distribution

In previous studies of recombination in male mice [11], we identified strain-specific differences in the number of foci per cell of the DNA mismatch repair protein MLH1, known to mark the vast majority of sites of crossing-over [9,26,27]. We decided to exploit these differences to investigate the basis of the variation. Accordingly, we examined three inbred strains -C57BL/6J ("B6"), CAST/Ei ("CAST") and C3H/HeJ ("C3H") – assaying a minimum of 15 pachytene stage cells per mouse, and at least five mice per strain, scoring the number of autosomal MLH1 foci per cell (Figure 1A).

Two of the inbred strains, CAST and B6, had previously been found to have "low" and "high" genome-wide MLH1 values, respectively [11]. Our re-analysis produced virtually identical results: the mean number +/- S.D. of autosomal MLH1 foci per cell was 21.3+/-1.6 for CAST (n = 105 cells) and 25.0+/-2.2 for B6 (n = 102 cells) (Figure 1B; Table S1). Subsequently, we analyzed C3H males, and observed that this strain had mean MLH1 values that were intermediate to the other two strains: i.e., 22.7+/-1.9 (n = 209 cells) (Figure 1B; Table S1). Because the number of MLH1 foci per cell is not normally distributed (i.e., typically each bivalent has at least one focus, thus constraining the autosomal foci per cell to 19 or more), inter-strain differences have to be interpreted with caution. Nevertheless, ANOVA analyses are typically robust in the face of modest departures from normality, and the magnitude of the differences we observed (F = 105.1; p< 0.0001), make it likely that the variation is real. Thus, these observations confirm our previous conclusion of variation in genome-wide MLH1 values - and presumably MLH1-driven crossovers - among males of different mouse strains.

The variation in overall MLH1 frequency was reflected by highly significant strain-specific differences in the proportion of chromosomes with zero, one, two, or three MLH1 foci $(\chi^2 = 292.0, p < 0.00001;$ Table S1). In large part, the difference was attributable to differing ratios of chromosomes with one vs. two MLH1 foci in the three strains; i.e., 6.8:1, 3.9:1 and 2.2:1 for CAST, C3H and B6, respectively (Table S1). However, intriguingly, the strain with the highest MLH1 average value (B6), also had the highest proportion of bivalents lacking MLH1 foci; indeed, this value was significantly increased over that for CAST ($\chi^2 = 12.1$, p<0.001) and for C3H ($\chi^2 = 13.9$, p<0.001) males.

Subsequently, we analyzed the placement of MLH1 foci among the three strains, asking whether variation in MLH1 levels might be linked to differences in the location of the foci. Initially, we simply pooled results from all homologs and found no obvious differences among the strains; i.e., consistent with previous results, distally located MLH1 foci predominated among all three strains [11]. However, because the strain-specific differences in the proportion of homologs with one, two and three MLH1 foci complicate interpretations of these data, we conducted a second set of studies in which we analyzed individual chromosomes. Specifically, for each strain, we analyzed the placement of MLH1 foci on chromosomes 19 and 1 on SCs that exhibited one and two foci, respectively (Figure 1C, D). No obvious differences were observed among the strains.



Figure 1. Inter-strain variation in mean MLH1 values. (A) Pachytene cell from B6 male immunostained with antibodies to MLH1 (green) and SYCP3 (red). The number of MLH1 foci per cell were counted and used as a surrogate for meiotic recombination events. (**B**) The mean number of MLH1 foci per spermatocyte for 5 CAST (n of cells = 105), 7 C3H (n of cells = 209) and six B6 (n of cells = 110) males varied significantly among the three inbred strains. (**C**, **D**) For each strain, slides were hybridized with FISH probes for chromosomes 19 and 1 and the positions of MLH1 foci were calculated as a percent of SC length (centromer = 0%; telomer = 100%). (**C**) On chromosomes 19 with a single MLH1 focus, the focus was typically medially or distally placed; data are based on analysis of 50 chromosomes/strain. (**D**) On chromosomes 1 with two MLH1 foci, typically one was proximally located and the other distally placed, consistent with positive interference. Data based on analysis of 50 cells/strain. doi:10.1371/journal.pgen.1004125.q001

Localization patterns of proteins involved in repair of DSBs vary among strains

The MLH1 data demonstrate inter-strain differences in recombination, but provide no information on when the variation originates. To address this question, we examined the abundance and distribution of signals for proteins involved at different stages of DSB repair, as follows:

Single end invasion is the earliest stage of recombination that can be consistently assayed using immunofluorescence. The ubiquitous RAD51 protein forms a complex around the resected ends of DSBs and facilitates invasion of the intact homologous chromosome [28]. We assayed localization patterns of RAD51 foci on zygotene stage chromosomes (Figure 2A) when RAD51 activity is high [29]; at this stage most RAD51 foci localize to synapsed regions, although in a few instances they (and also DMC1 foci, see below) are associated with unsynapsed axial elements. From Figure 2A, it is clear that there was considerable among-animal variation, particularly in the C3H and B6 strains. Nevertheless, considering the pooled data, the mean numbers of RAD51 foci per cell were highly significantly different among the three strains (F = 113.7; p<0.0001), with the variation mirroring that observed for MLH1; i.e., CAST had the lowest mean value (163.0+/-18.6), C3H an intermediate value (179.9+/-28.0), and B6 the highest value (222.1+/-33.8) (Figure 2A; Table S2). Further, considering those animals for which we had both RAD51 and MLH1 values (see Tables S1 and S2), the ratios of RAD51:MLH1 foci were similar among the three strains, with values of 7.7:1 for CAST, 7.3:1 for C3H and 8.8:1 for B6.

Other than RAD51, relatively few animals were scored for both MLH1 and another recombination protein (i.e., either DMC1, MSH4 or BLM) in each of the three strains; thus, we were not able to directly compare the ratios of MLH1:DMC1, MLH1:MSH4 or MLH1:BLM among the three strains. Nevertheless, the pooled data were consistent with the results from MLH1 and RAD51. For example, similar strain-specific differences were observed for zygotene stage cells scored for the meiosis specific strand invasion



Figure 2. Inter-strain variation in early- and mid-stage meiotic recombination proteins. The number and location of foci for proteins acting upstream of MLH1 in the recombination pathway [i.e., (A) RAD51 (B) DMC1 (C) MSH4] or anti-recombination pathway [i.e., (D) BLM] were determined for each of the three strains and inter-strain values compared. For each of the four proteins, mean numbers of foci per cell varied significantly among the three strains. Data for each protein were based on 4-6 animals per strain, and a minimum of 60 cells per strain (see Supplemental Tables 2-5 for data).

doi:10.1371/journal.pgen.1004125.g002

protein DMC1, which attaches to DSB sites shortly after RAD51 [29,30]. Specifically, the mean values rose from 119.5 ± 7.64 for CAST to 149.3+/-18.1 for C3H and 181.8+/-21.1 for B6 (F = 214.3; p < 0.0001) (Figure 2B; Table S3). Consistent with previous reports on DMC1 localization in mice [29], all three strains exhibited fewer DMC1 than RAD51 foci. While the reason for this variation in numbers of DMC1 and RAD51 foci is not entirely clear, it presumably reflects the different roles the two proteins play in the early stages of the recombination pathway [28,31,32] or possibly, differences in the time that the proteins remain complexed to the recombination nodules.

We investigated a later stage of recombination by assaying MSH4, a member of the MSH4-MSH5 complex that is thought to stabilize recombination intermediates (e.g., [19,22,33]. Since MSH4 localizes only to synapsed chromosome regions (Figure 2C), we counted foci in cells at the zygotene/pachytene boundary. We observed substantial variation in the values among the different individual mice but, similar to the results for RAD51 and DMC1, we saw an increase in the mean number of MSH4 foci from CAST (105.3 + / - 12.8) to C3H (109.0 + / - 14.8) and B6 (112.7+/-17.5) males (Figure 2C; Table S4). These differences were more modest than those identified for RAD51 and DMC1 and because there was considerable overlap in individual values from strain to strain, they need to be interpreted with caution. Nevertheless, the mean values for the "low" (CAST) and "high" (B6) strains were still highly significantly different from one another (t = 2.9; p< 0.01); for CAST vs C3H and C3H vs B6 the values were non-significantly different (t = 1.5, p > 0.13 and t = 1.5, p > 0.14, respectively)

Finally, we analyzed zygotene cells for the helicase BLM, a regulator of recombination intermediates suggested to have anticrossover activity ([23]; Figure 2D; Table S5). Strain specific differences in average numbers of BLM foci were similar to those observed for the positive regulators of crossovers; i.e., the mean value was lowest for CAST (114.4 + / -15.9), intermediate for C3H (120.0+/-12.4), and highest for B6 (146.4+/-26.3) (F = 62.3; p< 0.0001).

In summary, our observations on strain-specific variation for five different recombination pathway proteins (RAD51, DMC1, MSH4, BLM and MLH1) are consistent with one another; i.e., in each instance the mean number of foci per cell was lowest for CAST males, intermediate for C3H males, and highest for B6 males. Unfortunately, we were not able to collect data on each protein from each animal, limiting our ability to directly compare values among the different strains and, accordingly, to address other obvious questions, such as: which protein is the best predictor of MLH1 values, are the ratios of foci for different proteins the same among all three strains, and does the ratio of recombinogenic:antirecombinogenic proteins (e.g., RAD51:BLM) vary among strains? Nevertheless, taken together, the data provide strong evidence that, at least in males of these strains, a similar proportion of DSBs are translated into MLH1-associated crossovers.

Variation in chromatin configuration is correlated with MLH1 levels

In subsequent studies, we were interested in determining whether the strain differences were accompanied by variation in the configuration of the meiotic axis and/or DNA loops. Accordingly, for each strain we assayed total autosomal SC lengths in pachytene stage cells. We observed highly significant differences in mean SC length among the strains; i.e., for CAST 156.7 + / - 2.0 µm, for C3H 161.5 + / - 1.8 µm, and for B6 $170.7 + / -1.9 \ \mu m$ (F = 13.8; p<0.0001). These results are similar to the strain-specific observations for MLH1, suggesting that the strain differences in crossover levels are linked to variation in length of the meiotic axis (Figure 3A).

As a surrogate for DNA loop size, we examined the width of the signal from whole chromosome paint probes on representative large, medium-sized and small chromosomes; i.e., chromosomes, 1, 12 and 19, respectively (see Figure 3B for an example of a chromosome 12 paint probe). DNA loop sizes differed significantly among the strains for each of the three chromosomes: i.e., for chromosome 1, $CAST = 6.5 + / -1.3 \mu m$, $C3H = 4.5 + / -0.6 \mu m$ and B6 = 3.8 + / - $0.7 \ \mu m (F = 183.5; p < 0.0001);$ for chromosome 12, CAST = 6.2 + / -1.2 μ m, C3H = 4.3+/-0.6 μ m and B6 = 3.6+/-0.7 μ m (F = 176.5; p < 0.0001; and for chromosome 19, CAST = 4.4+/-1.0 µm, $C3H = 3.0 + / -0.5 \ \mu m$ and $B6 = 2.7 + / -0.5 \ \mu m$ (F = 116.5; p < 0.0001) (Figure 3C). Together with the observations on SC length, these analyses indicate that increasing MLH1 values are associated with smaller DNA loops and longer SCs.

DSB number per se does not regulate MLH1 levels or chromatin morphology

The correlation between the number of foci for "early", "mid", and "late" recombination proteins and strain-specific recombination levels raises the possibility that the number of DSBs per se regulates recombination. We tested this by comparing meiotic profiles of males on the same genetic background but with different rates of DSB formation. Specifically, we compared wildtype B6 males with siblings heterozygous for a null allele of Spol1, the type II topoisomerase-like protein responsible for programmed DSB formation in meiocytes [24]. Notably, the results from the wildtype Spo11 males were somewhat different than those of the B6 male described above, presumably due to variation from maintenance of the stocks at different facilities.

For the Spol1 animals, consistent with previous reports [34,35], zygotene spermatocytes from Spo11^{+/-} males displayed significantly fewer DSBs (estimated by the number of RAD51 foci) than spermatocytes from wildtype littermates (mean values = 152.2 + / -20.6 and 200.5 ± -21.5 , respectively; t = 5.8, p < 0.0001) (Figure 4A;Table S6). However, this did not translate into a difference in MLH1 values, with mean values of 23.9+/-1.9 and 23.8+/-1.6 for heterozygotes and wildtype littermates, respectively (Figure 4B; Table S7).

Although it altered RAD51 values, Spol1 heterozygosity had no obvious effect on chromatin morphology. Specifically, total SC lengths per cell were virtually identical for Spo11^{+/+} and Spo11^{+/-} littermates $(167.1 + / - 3.3 \ \mu m \text{ and } 169.7 + / - 4.8 \ \mu m, \text{ respectively;}$ t = 0.46 p > 0.65) (Figure 4C). Further, there were no consistent differences in DNA loop size on individual chromosomes between $Spo11^{+/+}$ and $Spo11^{+/-}$ males [i.e, $3.3+/-0.5 \ \mu m$ and 3.2+/- $0.5 \mu m$, respectively, for chromosome 1 (t = 1.01, p = 0.31); $2.9+/-0.5 \ \mu m$ and $2.9+/-0.5 \ \mu m$, respectively, for chromosome 12 (t = 0.27, p = 0.78), and 2.8+/-0.3 μ m and 2.7+/-0.4 μ m,



Figure 3. Inter-strain variation in SC length and DNA loop size. (**A**) Mean autosomal SC lengths varied in direct relationship to MLH1 values; i.e., the strain with the lowest mean MLH1 values (CAST) had the shortest SCs, while the strain with the highest MLH1 values (B6) had the longest SCs. Data are based on analyses of mid-pachytene spermatocytes from 2 CAST (n = 8 cells), 3 C3H (n = 11 cells) and 2 B6 (n = 21 cells) males. (**B**) DNA loop sizes for individual chromosomes were calculated as the average of the width of the FISH signal at three points along the SC: at the centromere, the midpoint, and the telomere; image shows example for chromosome 12. (**C**) Chromosomes 1, 12, and 19 were examined and for each, we observed an inverse relationship between strain-specific mean MLH1 values and mean DNA loop sizes.

respectively, for chromosome 19 (t = 1.00, p = 0.32)] (Figure 4D). Taken together with the results from the RAD51 and MLH1 assays, this indicates that the variation in crossover level among strains is not simply due to variation in the number of DSBs but more likely reflects differences in chromatin morphology.

Discussion

The purpose of this study was to investigate the basis for individual variation in recombination rates. Over the past 10–15 years a number of tools have become available to investigate the biology of meiotic recombination in mammals; e.g., knockout mice have been used to identify and characterize the functions of numerous meiotic genes (e.g., [20,36,37,38,39]); genotypic analysis of individual and pooled sperm samples has led to the identification of small, discrete regions of high recombination activity [40,41,42,43]; linkage and linkage disequilibrium studies have revealed the presence of thousands of recombination hotspots in mammalian genomes [7,44,45]; and genome-wide analyses of genetic polymorphisms have led to the identification of genes involved in modulating hotspot activity [46,47,48] or genomewide recombination levels [13,18].

Nevertheless, our understanding of the origin of amongindividual variation in recombination levels in mammals remains rudimentary. In recent studies, genotypic differences at three loci have been linked to variation in the recombination phenotype. Specifically, allelic variation in the gene encoding the meiosisspecific histone methytransferase PRDM9 has been associated with hotspot activity in both mice and humans [46,47,48]; allelic variation in RNF212, a homolog of the C. elegans



Figure 4. *Spo11*^{+/+} **and** *Spo11*^{+/-} **animals.** (**A**) Mice heterozygous for a null allele of *Spo11* exhibited a significant decrease in RAD51 foci (a marker of DSBs) by comparison to wildtype littermates. However, (**B**) the mean number of MLH1 foci (a marker of COs), (**C**) mean SC lengths and (**D**) mean DNA loop sizes were not different between the two genotypes. doi:10.1371/journal.pgen.1004125.g004

synapsis/cross-over associated gene zhp-3, has been shown to affect genome-wide recombination levels in human males and females [13,14,18]; and the presence of an inversion at 17q21.31 affects recombination rates in human females [13,14,49]. However, several lines of evidence indicate that these individual loci may be relatively unimportant, or at least not the only, determinants of variation in CO levels. First, in studies of PRDM9 in humans, allelic variation has been shown to influence hotspot usage but appears to have relatively little effect on genome-wide recombination levels [8,46,47,50]. Second, the magnitude of the effects attributable to the other two loci (RNF212 and inv17q21.31) is insufficient to account for the level of variation in genome-wide rates identified in humans or in the present study [13,18]. Finally, in recent analyses of inbred strains of mice with differing levels of genome-wide recombination conducted by us [51] and others [52], putative QTL-containing chromosome regions did not include either Prdm9 or Rnf212. Thus, the available evidence suggests that other, as yet unidentified, loci, are responsible for generating most of the variation in recombination rates among individuals or inbred strains.

To address this problem in a somewhat different way, we utilized inbred strains of mice with varying levels of genome-wide recombination to identify the temporal window during which the variation is generated. Surprisingly, our results provide strong evidence that the variation is attributable to processes acting at, or upstream of, DSB formation and repair. Specifically, our observations on foci number for each of three recombinationpromoting proteins (RAD51, DMC1, and MSH4) and for the anti-recombination protein BLM were similar in finding a direct correlation between foci numbers and levels of recombination. In each of the three strains, the number of DSBs (measured as RAD51 foci) exceeded the number of crossovers (MLH1 foci) by approximately ten-fold. This is consistent with observations from previous studies of male mice (e.g, [53]) and indicates that as in other organisms, the vast majority of DSBs are repaired as noncrossovers. Further, it suggests that, at least for the inbred strains we examined, events occurring downstream of DSBs are relatively unimportant in mediating genetic background-dependent variation in recombination rates. This does not mean that these processes are irrelevant - e.g., RNF212, which is known to affect

recombination levels in humans, has recently been show to stabilize proteins important in processing recombination intermediates in male mice [19] – only that there are other, more important determinants.

While our observations indicate that the variation in recombination levels is established by the time of DSB formation, our analyses of Spol1 deficient animals suggest that this is not attributable to DSBs per se. That is, despite having significantly different numbers of RAD51 foci, wildtype and heterozygous animals had virtually identical mean MLH1 values. These results are similar to those recently reported by Cole and colleagues [17], in which they demonstrated a positive correlation between the number of functional *Spol1* genes and the number of DMC1 and RAD51 foci, but no accompanying change in the number of MLH1 foci. Thus, as in other model organisms (e.g., [16,54]), mammalian meiosis appears to have homeostatic controls that operate to maintain the number of Cossovers in the face of variation in the number of DSBs.

Taken together, our analyses of different inbred strains of mice and of mice varying at a single locus but on an otherwise isogenic background demonstrate that DSB number itself is not the driver of variation in CO levels. What other processes might be responsible? One obvious possibility is the way in which the chromatin is packaged, an idea consistent with the observations of the present report. For example, the best predictor of the number of MLH1 foci across the five different categories of mice that we examined (i.e., CAST, C3H, B6, Spo11^{+/+} and Spo11^{+/-}) was SC length, since in each instance there was an average of approximately 7 µm SC per MLH1 focus. Similarly, the relationship between DNA loop sizes on individual chromosomes and the number of MLH1 foci was comparable in animals of different genetic background or Spol1 genotype. An obvious caveat to this interpretation is the fact that the observations on SC length and DNA loop size were based on pachytene cells while, presumably, the determination of the relative level of COs occurs much earlier. However, in unrelated studies of human males and females, we found that sex-specific differences in chromatin morphology in pachytene stage meiocytes were mirrored in leptotene preparations [55]. Thus, we think it unlikely that the observations of the present report simply reflect pachytene cells. Accordingly, we suggest that modification of chromatin morphology - but not DSB numbers - is a primary determinant of CO levels. This is consistent with recent analyses of Petukhova and Camerini-Otero and colleagues [56,57], as they found that trimethylation of H3K4 at potential PRDM9 recombination hotspots is not dependent on SPO11; i.e., the establishment of hotspot-associated marks occurs regardless of the presence of DSBs.

What genetic loci might be at work to influence chromatin packaging and, ultimately, variation in CO levels? In two recent QTL analyses of recombination in different inbred strains of [51,52], several QTLs were shared between the studies; i.e., overlapping regions were identified on chromosomes 4, 15 and 17, and nearly overlapping regions on chromosome 3 and the X chromosome. Notably, in both studies the highest lod scores were observed on the X chromosome, consistent with previous studies linking X-linked loci with variation in genome-wide recombination rates [12,58]. Because inactivation of the sex chromosomes (MSCI) appears to be essential for normal male meiosis [59], it is generally thought that few, if any, X-linked loci are necessary for spermatogenesis. The results of the analyses of Murdoch et al [51] and Dumont and Payseur [52], as well as studies identifying spermatogenic functions for X-linked genes (e.g., Tex11; [60]), suggests that this is not an absolute rule, and that high resolution analyses of the X chromosome may uncover important recombination-associated loci.

Materials and Methods

Ethics statement

All animal experiments and procedures conducted at Washington State University were performed using protocols approved by the Institutional Animal Care and Use Committee (WSU IACUC number 03737-014).

Mice

Breeding stocks of three inbred strains, C57BL/6J, C3H/HeJ, and CAST/EiJ, were obtained from The Jackson Laboratory and maintained by brother-sister mating. *Spo11*^{+/+} and *Spo11*^{+/-} animals (on a C57BL/6J background) were kindly provided by Drs. Maria Jasin and Scott Keeney, Memorial Sloan Kettering Cancer Center. All animals were housed in polysulfone cages on ventilated racks or static cages. Experiments were approved by the Washington State University (WSU) Institutional Animal Care and Use Committee. WSU is fully accredited by the American Association for Accreditation of Laboratory Animal Care and all investigations were conducted in accordance with the Guide for the Care and Use of Laboratory Animals.

Tissue collection

Adult male mice between the ages of 6 and 40 weeks were killed by cervical dislocation and the testes removed. Seminiferous tubules were removed from the testes and surface spread preparations of spermatocytes for immunofluorescence studies prepared as described previously [61].

Immunostaining

Slides were immunostained using similar methodology to that of Anderson *et al.* [26]. Antibodies were diluted in sterile filtered $1 \times ADB$ consisting of 10 ml normal donkey serum (Jackson ImmunoResearch), 3 g BSA (Sigma-Aldrich), 50 µl Triton X-100 (Alfa Aesar), and 990 ml PBS. Incubations were performed in a dark humid chamber at 37°C.

Slides were incubated overnight in a dark humid chamber at 37°C with one of the following primary antibodies: RAD51 (rabbit anti-human, Santa Cruz Biotechnology) diluted 1:75, DMC1 (rabbit anti-human, Santa Cruz Biotechnology) diluted 1:200, BLM (rabbit anti-human, Santa Cruz Biotechnology) diluted 1:100, MSH4 (rabbit anti-human, provided by Dr. Chengtao Her) diluted 1:75, or MLH1 (rabbit anti-human, Calbiochem) diluted 1:75. All slides were incubated with a primary antibody for SYCP3 (goat anti-mouse, Santa Cruz Biotechnology) diluted 1:100 for 2 hours. Fluorescein-labeled donkey anti-rabbit (Jackson ImmunoResearch) secondary antibody was diluted 1:75 and incubated overnight, followed by an incubation for 45 minutes with rhodamine-labeled donkey anti-goat secondary antibodies (Jackson ImmunoResearch), diluted 1:200. Slides were counterstained and fixed using Prolong Gold antifade reagent with DAPI (Invitrogen) and sealed with rubber cement.

Fluorescence in-situ hybridization

StarFish whole chromosome paint probes (Cambio) were used according to the manufacturer's instructions to estimate DNA loops of specific chromosomes. Briefly, previously immunostained slides were dehydrated in a series of brief ethanol washes (75%, 90%, 100%), denatured in a 70% dionized formamide/2×SSC solution for 5 minutes at 65°C, quenched in 70% ethanol at – 20°C, and dehydrated again with a series of ethanol washes. Paint probes for chromosomes 1, 12, and 19 were denatured for 10 minutes at 65°C and applied to the slides overnight in a dark humid chamber at 37°C. Following incubation, slides were soaked twice for 5 minutes in 50% dionized formamide/ $1 \times$ SSC solution at 45°C, washed twice in $1 \times$ SSC at 45°C for 5 minutes, and soaked for three cycles in 25% Detergent DT/ $4 \times$ SSC (Cambio) for 4 minutes at 45°C. Slides were allowed to drain and mounted with Reagent MD (Cambio) and sealed with rubber cement.

Microscopy and scoring

For immunofluorescence analysis, cells were identified using DAPI and the sub-stage of meiotic prophase determined on the basis of the morphology of the SC (visualized with SYCP3 antibody). For each cell and each protein analyzed, we scored the number of foci localizing to the axial element or SC at the appropriate sub-stage of meiotic prophase; i.e., for RAD51, DMC1, and BLM prior to complete synapsis during zygotene, MSH4 at the zygotene-pachytene transition, and MLH1 during pachytene prior to desynapsis of the XY bivalent. For analysis of MLH1 foci we restricted our scoring to autosomal chromosomes, since the appearance and disappearance of foci on the XY bivalent and on the autosomes are temporally uncoupled.

Position data for MLH1 foci was collected by measuring along the fully synapsed SC from the centromere to each MLH1 focus. The placement of MLH1 was calculated by dividing the centromere-MLH1 distances by the full length of each individual SC. Values are represented as a percentage of total SC length, with 0% being the centromeric end and 100% the telomeric end of the SC.

Genome-wide SC length was measured in fully synapsed pachytene cells by manually tracing the length of the SYCP3 signal for all SCs except the XY bivalent. The genome-wide SC length per cell was calculated as the sum of the SYCP3 signal lengths for the 19 autosomes.

DNA loop sizes were assayed on three representative chromosomes, 1, 12, and 19, identified by FISH paint probes, using an approach similar to that previously described by Novak et al [62] FISH images were then overlaid on immunofluorescence images of the SC and loop. Loop size was assayed by taking linear measurements of the width of the FISH signal perpendicular to the SC at the centromere, telomere, and mid-point of the SC. The three lengths for each chromosome were averaged and compared on a chromosome-specific basis among the strains.

All slides were imaged on a Zeiss Axio Imager epifluorescence microscope and analyzed by blinded observers using Zeiss Axiovision software (version 4.7).

Statistical analysis

Comparisons in the average numbers of foci between different strains or genotypes of mice were tested by analysis of variance or Student t-test analysis, pooling the results from multiple mice in a strain or genotype. For post-hoc comparisons of mean values

References

- Rockmill B, Roeder GS (1994) The yeast medl mutant undergoes both meiotic homolog nondisjunction and precocious separation of sister chromatids. Genetics 136: 65–74.
- Molnar M, Parisi S, Kakihara Y, Nojima H, Yamamoto A, et al. (2001) Characterization of rec7, an early meiotic recombination gene in Schizosaccharomyces pombe. Genetics 157: 519–532.
- Zetka MC, Rose AM (1995) Mutant rec-1 eliminates the meiotic pattern of crossing over in Caenorhabditis elegans. Genetics 141: 1339–1349.
- Hawley RS, Theurkauf WE (1993) Requiem for distributive segregation: achiasmate segregation in Drosophila females. Trends Genet 9: 310–317.
- Hassold T, Hunt P (2001) To err (meiotically) is human: the genesis of human aneuploidy. Nat Rev Genet 2: 280–291.
- Cheung VG, Burdick JT, Hirschmann D, Morley M (2007) Polymorphic variation in human meiotic recombination. American journal of human genetics 80: 526–530.
- Coop G, Wen X, Ober C, Pritchard JK, Przeworski M (2008) High-resolution mapping of crossovers reveals extensive variation in fine-scale recombination patterns among humans. Science 319: 1395–1398.

between specific strains, we used the Bonferroni correction to account for the multiple tests. Mean values +/- SD are provided in the text and tables and, for illustrative purposes, as mean values +/- SE in the figures. Differences in the numbers of E0 bivalents were compared by chi-square tests. All statistical analyses were performed with JMP software, version 7.0.1 (SAS Institute Inc.).

Supporting Information

Table S1 Mean +/- S.D. MLH1 foci numbers for each animal and inbred strain. Significant strain-specific differences were evident, with B6 having the highest mean values, C3H intermediate values and CAST the lowest values. Similar strainspecific differences were observed for other recombination pathway proteins, and are summarized below in Tables S2-S5. (DOCX)

Table S2Mean +/- S.D. RAD51 foci numbers for each animaland inbred strain.

(DOCX)

Table S3Mean +/ - S.D. DMC1 foci numbers for each animaland inbred strain.

(DOCX)

Table S4Mean +/- S.D. MSH4 foci numbers for each animaland inbred strain.

(DOCX)

Table S5 Mean +/- S.D. BLM foci numbers for each animaland inbred strain.

(DOCX)

Table S6 Mean +/- S.D. RAD51 foci numbers for *Spo11*^{+/-} and *Spo11*^{+/+} males. (DOCX)

Table S7 Mean +/- S.D. MLH1 foci numbers for $Spoll^{+/-}$ and $Spoll^{+/+}$ males. (DOCX)

Acknowledgments

We gratefully acknowledge the assistance of Crystal Lawson in the preparation of this manuscript.

Author Contributions

Conceived and designed the experiments: BB PH TH. Performed the experiments: BB TH. Analyzed the data: BB PH KWB TH. Contributed reagents/materials/analysis tools: BB PH TH. Wrote the paper: BB PH KWB TH.

- Kong A, Thorleifsson G, Gudbjartsson DF, Masson G, Sigurdsson A, et al. (2010) Fine-scale recombination rate differences between sexes, populations and individuals. Nature 467: 1099–1103.
- Lynn A, Kochler KE, Judis L, Chan ER, Cherry JP, et al. (2002) Covariation of synaptonemal complex length and mammalian meiotic exchange rates. Science 296: 2222–2225.
- Hassold T, Hansen T, Hunt P, VandeVoort C (2009) Cytological studies of recombination in rhesus males. Cytogenetic and genome research 124: 132–138.
- Koehler KE, Cherry JP, Lynn A, Hunt PA, Hassold TJ (2002) Genetic control of mammalian meiotic recombination. I. Variation in exchange frequencies among males from inbred mouse strains. Genetics 162: 297–306.
- Dumont BL, White MA, Steffy B, Wiltshire T, Payseur BA (2011) Extensive recombination rate variation in the house mouse species complex inferred from genetic linkage maps. Genome research 21: 114–125.
- Chowdhury R, Bois PR, Feingold E, Sherman SL, Cheung VG (2009) Genetic analysis of variation in human meiotic recombination. PLoS genetics 5: e1000648.

- Fledel-Alon A, Leffler EM, Guan Y, Stephens M, Coop G, et al. (2011) Variation in human recombination rates and its genetic determinants. PloS one 6: e20321.
- Matise TC, Chen F, Chen W, De La Vega FM, Hansen M, et al. (2007) A second-generation combined linkage physical map of the human genome. Genome Res 17: 1783–1786.
- Martini E, Diaz RL, Hunter N, Keeney S (2006) Crossover homeostasis in yeast meiosis. Cell 126: 285–295.
- Cole F, Kauppi L, Lange J, Roig I, Wang R, et al. (2012) Homeostatic control of recombination is implemented progressively in mouse meiosis. Nature cell biology 14: 424–430.
- Kong A, Thorleifsson G, Stefansson H, Masson G, Helgason A, et al. (2008) Sequence variants in the RNF212 gene associate with genome-wide recombination rate. Science 319: 1398–1401.
- Reynolds A, Qiao H, Yang Y, Chen JK, Jackson N, et al. (2013) RNF212 is a dosage-sensitive regulator of crossing-over during mammalian meiosis. Nature Genetics 45: 269–278.
- Baker SM, Plug AW, Prolla TA, Bronner CE, Harris AC, et al. (1996) Involvement of mouse Mlh1 in DNA mismatch repair and meiotic crossing over [see comments]. Nat Genet 13: 336–342.
- Masson JY, West SC (2001) The Rad51 and Dmc1 recombinases: a nonidentical twin relationship. Trends in biochemical sciences 26: 131–136.
- Santucci-Darmanin S, Walpita D, Lespinasse F, Desnuelle C, Ashley T, et al. (2000) MSH4 acts in conjunction with MLH1 during mammalian meiosis. FASEB journal : official publication of the Federation of American Societies for Experimental Biology 14: 1539–1547.
- Jessop L, Rockmill B, Roeder GS, Lichten M (2006) Meiotic chromosome synapsis-promoting proteins antagonize the anti-crossover activity of sgs1. PLoS genetics 2: e155.
- Keeney S, Giroux CN, Kleckner N (1997) Meiosis-specific DNA double-strand breaks are catalyzed by Spol1, a member of a widely conserved protein family. Cell 88: 375–384.
- Keeney S, Baudat F, Angeles M, Zhou ZH, Copeland NG, et al. (1999) A mouse homolog of the Saccharomyces cerevisiae meiotic recombination DNA transesterase Spo11p. Genomics 61: 170–182.
- Anderson LK, Reeves A, Webb LM, Ashley T (1999) Distribution of crossing over on mouse synaptonemal complexes using immunofluorescent localization of MLH1 protein. Genetics 151: 1569–1579.
- Svetlanov A, Baudat F, Cohen PE, de Massy B (2008) Distinct functions of MLH3 at recombination hot spots in the mouse. Genetics 178: 1937–1945.
- Sheridan SD, Yu X, Roth R, Heuser JE, Schorn MG, et al. (2008) A comparative analysis of Dmc1 and Rad51 nucleoprotein filaments. Nucleic acids research 36: 4057–4066.
- 29. Moens PB, Kolas NK, Tarsounas M, Marcon E, Cohen PE, et al. (2002) The time course and chromosomal localization of recombination-related proteins at meiosis in the mouse are compatible with models that can resolve the early DNA-DNA interactions without reciprocal recombination. Journal of cell science 115: 1611–1622.
- Tarsounas M, Morita T, Pearlman RE, Moens PB (1999) RAD51 and DMC1 form mixed complexes associated with mouse meiotic chromosome cores and synaptonemal complexes. J Cell Biol 147: 207–220.
- Barchi M, Mahadevaiah S, Di Giacomo M, Baudat F, de Rooij DG, et al. (2005) Surveillance of different recombination defects in mouse spermatocytes yields distinct responses despite elimination at an identical developmental stage. Molecular and cellular biology 25: 7203–7215.
- Callender TL, Hollingsworth NM (2010) Mek1 suppression of meiotic doublestrand break repair is specific to sister chromatids, chromosome autonomous and independent of Rec8 cohesin complexes. Genetics 185: 771–782.
- Borner GV, Kleckner N, Hunter N (2004) Crossover/noncrossover differentiation, synaptonemal complex formation, and regulatory surveillance at the leptotene/zygotene transition of meiosis. Cell 117: 29–45.
- Bellani MA, Romanienko PJ, Cairatti DA, Camerini-Otero RD (2005) SPO11 is required for sex-body formation, and Spo11 heterozygosity rescues the prophase arrest of Atm-/- spermatocytes. Journal of cell science 118: 3233–3245.
- Barchi M, Roig I, Di Giacomo M, de Rooij DG, Keeney S, et al. (2008) ATM promotes the obligate XY crossover and both crossover control and chromosome axis integrity on autosomes. PLoS genetics 4: e1000076.
- Baker SM, Bronner CE, Zhang L, Plug AW, Robatzek M, et al. (1995) Male mice defective in the DNA mismatch repair gene PMS2 exhibit abnormal chromosome synapsis in meiosis. Cell 82: 309–319.
- Yuan L, Liu JG, Hoja MR, Wilbertz J, Nordqvist K, et al. (2002) Female germ cell aneuploidy and embryo death in mice lacking the meiosis-specific protein SCP3. Science 296: 1115–1118.

- Baudat F, Manova K, Yuen JP, Jasin M, Keeney S (2000) Chromosome synapsis defects and sexually dimorphic meiotic progression in mice lacking spo11. Mol Cell 6: 989–998.
- Edelmann W, Cohen PE, Kneitz B, Winand N, Lia M, et al. (1999) Mammalian MutS homologue 5 is required for chromosome pairing in meiosis. Nature Genetics 21: 123–127.
- Badge RM, Yardley J, Jeffreys AJ, Armour JA (2000) Crossover breakpoint mapping identifies a subtelomeric hotspot for male meiotic recombination. Human Molecular Genetics 9: 1239–1244.
- Lien S, Szyda J, Schechinger B, Rappold G, Arnheim N (2000) Evidence for heterogeneity in recombination in the human pseudoautosomal region: high resolution analysis by sperm typing and radiation-hybrid mapping. American journal of human genetics 66: 557–566.
- Jeffreys AJ, Murray J, Neumann R (1998) High-resolution mapping of crossovers in human sperm defines a minisatellite-associated recombination hotspot. Mol Cell 2: 267–273.
- Jeffreys AJ, May CA (2004) Intense and highly localized gene conversion activity in human meiotic crossover hot spots. Nature Genetics 36: 151–156.
- McVean GA, Myers SR, Hunt S, Deloukas P, Bentley DR, et al. (2004) The fine-scale structure of recombination rate variation in the human genome. Science 304: 581–584.
- Myers S, Bottolo L, Freeman C, McVean G, Donnelly P (2005) A fine-scale map of recombination rates and hotspots across the human genome. Science 310: 321–324.
- Baudat F, Buard J, Grey C, Fledel-Alon A, Ober C, et al. (2010) PRDM9 is a major determinant of meiotic recombination hotspots in humans and mice. Science 327: 836–840.
- Berg IL, Neumann R, Lam KW, Sarbajna S, Odenthal-Hesse L, et al. (2010) PRDM9 variation strongly influences recombination hot-spot activity and meiotic instability in humans. Nature Genetics 42: 859–863.
- Parvanov ED, Petkov PM, Paigen K (2010) Prdm9 controls activation of mammalian recombination hotspots. Science 327: 835.
- Stefansson H, Helgason A, Thorleifsson G, Steinthorsdottir V, Masson G, et al. (2005) A common inversion under selection in Europeans. Nature Genetics 37: 129–137.
- Hinch AG, Tandon A, Patterson N, Song Y, Rohland N, et al. (2011) The landscape of recombination in African Americans. Nature 476: 170–175.
- Murdoch B, Owen N, Shirley S, Crumb S, Broman KW, et al. (2010) Multiple loci contribute to genome-wide recombination levels in male mice. Mammalian genome : official journal of the International Mammalian Genome Society 21: 550–555.
- Dumont BL, Payseur BA (2011) Genetic analysis of genome-scale recombination rate evolution in house mice. PLoS Genet 7: e1002116.
- Moens PB, Marcon E, Shore JS, Kochakpour N, Spyropoulos B (2007) Initiation and resolution of interhomolog connections: crossover and non-crossover sites along mouse synaptonemal complexes. Journal of cell science 120: 1017–1027.
- Rosu S, Libuda DE, Villeneuve AM (2011) Robust crossover assurance and regulated interhomolog access maintain meiotic crossover number. Science 334: 1286–1289.
- Gruhn J, Rubio C, Broman K, Hunt P, Hassold T (2013) Cytological studies of human meiosis: sex-specific differences in recombination originate at, or prior to, establishment of double-strand breaks. PloS one 8(12): e85075.
- Smagulova F, Gregoretti IV, Brick K, Khil P, Camerini-Otero RD, et al. (2011) Genome-wide analysis reveals novel molecular features of mouse recombination hotspots. Nature 472: 375–378.
- Brick K, Smagulova F, Khil P, Camerini-Otero RD, Petukhova GV (2012) Genetic recombination is directed away from functional genomic elements in mice. Nature 485: 642–645.
- de La Casa-Esperon E, Loredo-Osti JC, Pardo-Manuel de Villena F, Briscoe TL, Malette JM, et al. (2002) X chromosome effect on maternal recombination and meiotic drive in the mouse. Genetics 161: 1651–1659.
- Royo H, Polikiewicz G, Mahadevaiah SK, Prosser H, Mitchell M, et al. (2010) Evidence that meiotic sex chromosome inactivation is essential for male fertility. Current biology : CB 20: 2117–2123.
- Adelman CA, Petrini JH (2008) ZIP4H (TEX11) deficiency in the mouse impairs meiotic double strand break repair and the regulation of crossing over. PLoS genetics 4: e1000042.
- Peters AH, Plug AW, van Vugt MJ, de Boer P (1997) A drying-down technique for the spreading of mammalian meiocytes from the male and female germline. Chromosome Res 5: 66–68.
- Novak I, Wang H, Revenkova E, Jessberger R, Scherthan H, et al. (2008) Cohesin Smc1beta determines meiotic chromatin axis loop organization. The Journal of cell biology 180: 83–90.